Author’s response to reviews

Title: A case report of clonidine induced syncope: a review of central actions of an old cardiovascular drug

Authors:

Alexander Sandweiss (asand@email.arizona.edu)
Christopher Morrison (m0rris0n@email.arizona.edu)
Anne Spichler (annespichler@gmail.com)
John Rozich (johnrozich@hotmail.com)

Version: 2 Date: 23 Feb 2017

Author’s response to reviews:

Reviewer reports:

Mauro Cataldi, M.D., Ph.D. (Reviewer 1): The authors addressed satisfactorily all the points raised in the previous review.

Barbara Wiśniowska (Reviewer 2): The authors addressed all my comments satisfactorily.

Two minor additional comments:

- Please be consistent with ECG/EKG spelling
- Consider adding the drug name to the title

We have changed 'ECG' to 'EKG.' We have added clonidine to the title.